Author's response to reviews

Title: Increase in condom use and decline in prevalence of sexually transmitted infections among high-risk men who have sex with men and transgender persons in Maharashtra, India: Avahan, the India AIDS Initiative

Authors:

Shreena Ramanathan (sramanathan@fhi360.org)
Sucheta Deshpande (sucheta.ibba@gmail.com)
Abhishek Gautam (abhishekiips@gmail.com)
Dilip B Pardeshi (dbpardeshi@rediffmail.com)
Lakshmi Ramakrishnan (lramakrishnan@fhi360.org)
Prabuddhagopal Goswami (pgoswami@fhi360.org)
Rajatashvura Adhikary (RAdhikary@fhi360.org)
Bitra George (BGeorge@fhi360.org)
Ramesh S Paranjape (rparanjape@nariindia.org)
Mandar M Mainkar (mmainkar@nariindia.org)

Version: 2 Date: 21 June 2013

Author's response to reviews:

Date: 21 June 2013
To,
The Editor,
BMC-Public Health
Dear Editor,

We thank you for proving us an opportunity to resubmit our paper (MS: 1457241132880259) after modifications, based on comments from reviewers. The comments from reviewers were insightful and helped us address gaps in the manuscript. We have replied to each of the comments below and have incorporated the suggested revisions. We edited the complete manuscript to improve the flow and readability and sections with major changes have been highlighted.

We hope we have satisfactorily addressed the concerns of the reviewers, and the paper can now be considered for publication. Please let us know, if any further clarifications are needed.

Sincerely,

Shreena Ramanathan
Senior Technical Officer,
FHI 360 India,
H-5 Green Park Extension (Ground Floor),
New Delhi, India – 110016
Ph: 91-11-40487777
E-mail- sramanathan@fhi360.org